Author's response to reviews

Title: Improvements in survival of the uncemented Nottingham Total Shoulder prosthesis: a prospective comparative study

Authors:

Nahum Rosenberg (nahumrosenberg@hotmail.com)
Lars Neumann (larsneumann@rcsed.ac.uk)
Amit Modi (dramitmodi@yahoo.com)
Istvan J Mersich (istvan_mersich@hotmail.com)
Angus W Wallace (angus.wallace@rcsed.ac.uk)

Version: 3 Date: 31 July 2007

Author's response to reviews:

The Editor
BMC Musculoskeletal Disorders

Re: Manuscript submission

Dear Editor

Thank you for the acceptance of our paper "Improvements in survival of the uncemented Nottingham Total Shoulder prosthesis: a prospective comparative study" for publication in the BMC Musculoskeletal Disorders.

The following are the corrections in the manuscript we did according to your suggestions:

* The title is written in a sentence case
* * was added to the corresponding author name in the email author list
* The title page was aligned to the left
* Key words were removed
* Introduction title was changed to Background, List of abbreviations to Abbreviations
* In the tables the shaded areas were removed and bold fonts were used instead
* Figure 3 was changed to one a page long only

We hope that the paper is ready for publication

Yours Sincerely
Nahum Rosenberg MD, MOorth
Nottingham Shoulder and Elbow Unit
Nottingham City Hospital
Hucknall Road
Nottingham NG5 1PB
UK
Tel: + 44 (0)115 969 1169
Fax: + 44 (0)115 962 8062
Email: nahumrosenberg@hotmail.com